Author’s response to reviews

Title: Prevalence of Lower Extremity Peripheral Artery Disease among Adult Diabetic patients in Southwestern Uganda.

Authors:

Samson Okello (okello.samson@must.ac.ug)
Alexander Millard (millardunmc@gmail.com)
Rogers Owor (oworogers@hotmail.com)
Stephen B Asiimwe (asiimwesteve@gmail.com)
Mark J Siedner (msiedner@gmail.com)
Joselyn Rwebembera (jrwebembera@must.ac.ug)
Laurence A Wilson (tonywislon@gmail.com)
Christopher C Moore (CCM5U@hscmail.mcc.virginia.edu)
Brian H Annex (BHA4N@hscmail.mcc.virginia.edu)

Version: 2  Date: 5 May 2014

Author’s response to reviews:

We appreciate the reviewers’ constructive comments and feel that our responses have helped improve the quality and presentation of our manuscript. We provided, in the letter attached, a detailed point-by-point response to each of the reviewers’ queries.